Author's response to reviews

Title: Interest in neurology during medical clerkship in three Nigerian medical schools

Authors:

Emmanuel O Sanya (emanuelosanya@yahoo.com)
Olugbenga E Ayodele (gbengaayox@yahoo.ca)
Timothy Segun Olanrewaju (timothysegun@yahoo.com)

Version: 3 Date: 11 April 2010

Author's response to reviews:

The Editor 10/04/2010
BMC Medical Education
Article ID: 1365925278345391

Re: Resubmission of corrected article for consideration for publication now titled “Interest in Neurology during medical clerkship in three Nigerian medical schools”

I on behalf of other co-authors hereby resubmit the corrected version of our above titled article.

We like to thank the editors for their kind review of our article and given it a provisional approval for publication in your open access journal.

Ethical issues have been clearly stated in the methodology section that ethical approval was obtained from ethics committee of our institution and that each student gave verbal consent for participation in the study.

With this correction we hope our article can now be accepted fully for publication in your widely read open access journal.

Yours truly,
Emmanuel Sanya (MD).